Author's response to reviews

Title: Topiramate in the treatment of compulsive sexual behavior: case report

Authors:

Yasser Khazaal (yasser.khazaal@chuv.ch)
Daniele Zullino (daniele.zullino@hcuge.ch)

Version: 4 Date: 19 May 2006

Author's response to reviews:

Dear Editor,

-As suggested, the title of the manuscript was modified to:

-We add the following acknowledgement:
"The authors thank the patient for his agreement to this publication"

-We add the following sentence in the competing interests section
"No funding or support was received for the publication"

-We remove the reference to "Mr. A"

-Topiramate was capitalised, only when beginning a sentence.

-The recent article (Shiah IS 2006) was added to the references and to the background and discussion sections as follows:
"To our knowledge, the effectiveness of topiramate is reported in a paraphilic[22] as well as in a nonparaphilic sexual disorder [23]."

"While a spontaneous remission or a placebo effect cannot be completely ruled out in the present case, different criteria sustain topiramate efficacy: the previous case reports."..

-We remove the keywords section, as well as the following sentence: "To Biomed Central Psychiatry"

-We add the following sentence in the bottom of the background section:
"The authors obtained written consent from the patient for publication of this case report."

-Corrections were made to the "references section" as recommended.

Thank you